



# **A Comprehensive Review of Existing and Pending University Campus Microgrids**

Edrees Yahya Alhawsawi <sup>1,2</sup>, Khaled Salhein <sup>1</sup>, and Mohamed A. Zohdy <sup>1,\*</sup>

- <sup>1</sup> Department of Electrical and Computer Engineering, Oakland University, Rochester, MI 48309, USA; ealhawsawi@oakland.edu (E.Y.A.)
- <sup>2</sup> Department of Electrical and Computer Engineering, College of Engineering, Effat University, Jeddah 21478, Saudi Arabia
- \* Correspondence: zohdyma@oakland.edu

**Abstract:** Over the past few decades, many universities have turned to using microgrid systems because of their dependability, security, flexibility, and less reliance on the primary grid. Microgrids on campuses face challenges in the instability of power production due to meteorological conditions, as the output of renewable sources such as solar and wind power relies entirely on the weather and determining the optimal size of microgrids. Therefore, this paper comprehensively reviews the university campuses' microgrids. Some renewable energy sources, such as geothermal (GE), wind turbine (WT), and photovoltaic (PV), are compared in terms of installation costs, availability, weather conditions, efficiency, environmental impact, and maintenance. Furthermore, a description of microgrid systems and their components, including distributed generation (DG), energy storage system (ESS), and microgrid load, is presented. As a result, the most common optimization models for analyzing the performance of campus microgrids are discussed. Hybrid microgrid system configurations are introduced and compared to find the optimal configuration in terms of energy production and flexibility. Therefore, configuration A (Hybrid PV- grid-connected) is the most common configuration compared to the others due to its simplicity and free-charge operation.

Keywords: renewable-nonrenewable energy; microgrid; campus microgrid

# 1. Introduction

Adopting a utility power source is not an optimal solution due to its reliance on fossil fuels, which are both costly and detrimental to the environment. Hence, switching to renewable energy sources (RESs) is essential owing to their significant benefits [1]. RESs such as wind, solar, geothermal systems, biomass, tidal, and hydroelectricity are more effective, reliable, low maintenance, and offer efficiently lower carbon dioxide  $(CO_2)$  emissions and a quieter atmosphere [1-3]. In addition, RESs are cheaper than traditional resources in the long run, reducing energy costs while providing loads with dependable and sustainable energy. Furthermore, RESs can be integrated with nonrenewable energy sources, such as combined heat and power (CHP) and diesel generators (DGs), to fulfill the electricity demands and maximize power efficiency [4,5]. The advantages of RESs vary according to each source's conditions. For instance, geothermal energy (GE) is considered one of the most promising clean, renewable sources because it depends entirely on the Earth's heat to drive steam turbines that produce electricity. GE is available year-round and unaffected by weather and consequently stabilizes energy output, unlike other renewable energy sources, including wind, hydro, and solar panels [3,6]. Moreover, GE can be installed vertically almost everywhere, requiring less space than wind and solar. Accordingly, GE, wind, and solar panels each require approximately 404, 1335, and 2340 square miles of land surface to generate 1 gigawatt-hour (GWh) of electricity [7]. It is preferable for GE to be installed in an easily accessible area since it goes hundreds of meters below the surface of the Earth. The costs of producing 1 kilowatt (KW) utilizing geothermal, wind, and solar



Citation: Alhawsawi, E.Y.; Salhein, K.; Zohdy, M.A. A Comprehensive Review of Existing and Pending University Campus Microgrids. *Energies* 2024, *17*, 2425. https:// doi.org/10.3390/en17102425

Academic Editor: Tek Tjing Lie

Received: 20 April 2024 Revised: 9 May 2024 Accepted: 16 May 2024 Published: 18 May 2024



**Copyright:** © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). power are USD 3478, USD 1274, and USD 3.025, respectively [8–10]. In 2022, the worldwide installed capacity of wind, solar, and geothermal power reached 906 GWh, 710 GWh, and 14.9 GWh, respectively [8,10,11]. Due to the high cost of drilling boreholes and the high initial costs of facility construction, GE is a more costly power source compared to wind and solar [3,12,13]. Table 1 presents a comparison of renewable energy sources [14–24]. In summary, geothermal, wind, and solar power have similar characteristics, such as clean, sustainable energy sources and low environmental impact. Nevertheless, there are differences in terms of their availability, effectiveness, and installation expenses. Additionally, choosing renewable energy sources relies on such factors such as geographical location and energy demands.

Aspect	Solar Energy	Wind Turbines	Geothermal Energy
Availability	Depending on the location	Depending on the location	Worldwide
Efficiency	22%	20–40%	32%
Maintenance	Regular	Regular	Low
Installation cost	High	Low	High
Environmental impact (Greenhouse gas)	Low impact	Low impact	No impact
Weather conditions	Affected	Affected	Unaffected

Table 1. Comparison of renewable energy sources.

Hybrid power systems (HPSs) can produce electricity by combining two or more renewable and nonrenewable energy sources [25]. A hybrid energy source consists primarily of renewable energy sources, such as photovoltaic (PV), wind turbines (WT), and GE, and conventional energy sources, such as CHP, DG, and storage systems. Figure 1 shows the hybrid power systems. The HPS depends only on one power source to minimize costs and improve system efficiency. Hence, HPS is the most popular option used. There are some advantages of HPSs, such as providing energy to remote areas with high efficiency without the fixed source vulnerability associated with large-scale networks, which can lead to power grid failure [26,27]. In addition, the HPS reduces fuel costs while minimizing line losses and interruptions for consumers [28]. The HPS lowers emissions of pollutants and greenhouse gases [29].

A microgrid is a self-sufficient power grid that can operate either connected to the power grid or independently to provide electricity to various facilities, such as university campuses, commercial buildings, and hospital complexes. The islanded system is not connected to the primary grid. However, the grid-connected system is connected to the primary grid. Figure 2 shows a microgrid architecture. Lubna et al. [5] concluded that a microgrid system would be suitable for areas with inadequate transmission infrastructure, like isolated villages where an islanded microgrid would be the most beneficial type of power network. It is important to note that microgrids provide more dependable and secure energy sources as energy availability becomes increasingly subject to natural disasters and cyber-attacks [30–33]. Furthermore, microgrids are like traditional power grids regarding control, distribution, transmission, and power generation characteristics. The microgrid system, on the other hand, differs from conventional grids that can be installed near the load sites, thereby lowering the initial capital cost associated with the transmission lines between power generation and consumption cycles.

Microgrid systems have emerged as a sustainable and cost-effective solution for several university campuses. These systems are designed to make universities self-sufficient during load shedding and power outages [22]. Stephanus et al. [23] have highlighted the potential of microgrids in meeting the growing power needs of campuses while reducing operational utility expenses. The effectiveness of microgrid technology varies among universities, influenced by factors such as campus size, weather conditions, and

AC Sources Back Up Geothermal СНР Wind Turbne Diesel generato 0 Load

geographical location. Numerous studies have been conducted to enhance the overall campus microgrid's performance [34-36].

Figure 1. Hybrid power system.

Photovoltaic Diesel Generator (°) Utility Grid Microgrid Electric Vehicle Optimal Operation Combine Heat and Power Energy storge Geothermal energy Loads

Wind Turbine

Figure 2. Microgrid architecture.



3 of 29

This paper comprehensively reviews microgrid systems on university campuses, covering principles, types, and geographical locations using algorithms, connections, and applications. It also undertakes a comparative analysis of GE, WT, and PV in terms of installation cost, availability, weather conditions, efficiency, environmental impact, and maintenance. The paper introduces and compares hybrid microgrid system configurations, aiming to identify the optimal configuration for energy production and flexibility.

The rest of the paper is arranged as follows: Section 2 presents microgrid components, including distributed generation, energy storage system, and microgrid loads. Section 3 provides an overview of microgrids at different universities. Section 4 provides proposed optimization techniques. Section 5 presents an overview of campus microgrid architectures, including their configuration. Section 6 provides a brief conclusion.

# 2. Microgrid Components

The microgrid system has three main components: distributed generation, energy storage system, and loads. Figure 3 illustrates the microgrid components.



Figure 3. Microgrid components.

#### 2.1. Distributed Generation

Distributed generation (DG) is a method of generating electricity near the building where it is used, rather than sourcing electricity from traditional power plants. DG uses either conventional energy sources, including fuel cells (FCs), diesel, nuclear power, and natural gas, or renewable energy sources, including wind turbines (WT), biogas, geothermal energy (GE), and solar photovoltaic (PV) energy [37]. Some DG types can produce both combined heat and power by recovering some of the waste heat produced by the energy source. In addition, DG can lower energy costs and reduce greenhouse gas emissions. Consequently, this can dramatically enhance the DG unit's efficiency. Some distributed generation systems require a power electronics device (e.g., AC/DC converter) to convert the harvested energy to the utility grid. Distributed generation renewable sources are frequently utilized to generate electricity. The installed capacity of wind energy has increased significantly in the past decade, reaching almost 900 GW at the end of 2022 [38]. Along with solar PV energy, TW energy has become a significant microgrid resource. The world installed capacities for PV, WT, and GE are anticipated to be 2000 gigawatts, 1 terawatt, and 23.4368 gigawatts, respectively, by 2030 [1,39,40]. Furthermore, renewable energy is expected to contribute the most annual additions for the next decade relative to all fossil fuels, according to the International Energy Agency (IEA) [41]. It is crucial to match the system to the user's needs when using renewable energy sources for distributed power generation. The ratio of energy demand coverage and self-consumption rate are two important parameters that should be considered [42,43]. Figure 4 shows the distributed generation system.



Figure 4. Distributed generation system.

# 2.2. Energy Storage System

An energy storage system (ESS) is a way of storing power harvested from both renewable and non-renewable sources. This power can then be used for various valuable operations (e.g., homes, air conditioning, electronics, transportation, etc.) [44]. Since energy production does not match consumption due to factors such as energy demands and weather conditions, the ESS is the optimal solution to reduce imbalances between energy production and demands. In addition to reducing electricity costs, an ESS minimizes environmental impact, improves grid reliability, and allows the integration of diverse energy sources [45]. Harvested energy can be stored in various forms, such as electrochemical, thermal, hybrid, chemical, mechanical, and electrical [46]. Figure 5 shows the energy storage system classification. Some requirements of ESS components should be considered during microgrid design, such as balancing energy demand between the load and production. In addition, it is essential to store the highest energy capacity needs during off-peak hours and provide the energy demand when needed. Furthermore, smooth transient conditions from an islanded to a grid-connected microgrid and vice versa [4]. Table 2 illustrates the details of the ESS classification [47].



Figure 5. Classification of energy storage systems.

Ref.	Category	Types/Model	Operation	Modes	Advantages	Disadvantages
[48]	Electrochemical	Sodium sulphur, lead acid, nickel-cadium, and lithium-ion	Energy is converted from chemical to electrical energy in active materials	Batteries that are conventionally rechargeable, as well as batteries that are flow rechargeable	Storage devices come in a variety of sizes and require minimal maintenance	Chemical reactions reduce battery life and energy
[49,50]	Thermal	Solid (stone, concrete, metal, and ground), liquid with a solid filler material (molten salt/stone), or liquid (water, molten salt, and thermal oil)	Heat or ice is used to store energy	High-temperature and low-temperature	Technology is an alternative to fossil fuels that can meet the demands of sustainable energy laws. Provides a secure supply of energy, protects the environment, and achieves a high energy density	Low life expectancy
[51–55]	Hybrid	The battery can connect to an SC, an SMES, an FC, an SC, and an RFB.	Multi-ESS integration	Enhances the stability and reliability of the system while decreasing the problems associated with power quality by combining the characteristics of high power and high energy storage systems	Improves system efficiency and extends battery life	High costs
[56,57]	Chemical	Hydrogen, diesel, propane, ethanol, and liquefied petroleum gas	Electricity can be directly generated	Chemical bonds within atoms and molecules are responsible for storing energy	The availability of raw materials significantly reduces the cost per unit because they store significant amounts of energy for long periods	Developing this technology requires a high level of efficiency
[58,59]	Mechanical	Compressed air, flywheel, and pumped hydro storage	Assists mechanical work by delivering the stored power	Kinetic energy potential energy, forced spring, and pressurized gas	Utilizes flexible methods of converting and storing energy	Geologically, it is costly to implement, has a negative environmental impact, and is not economically feasible
[57,60]	Electrical	Super magnetic and supercapacitor	Electrical or magnetic fields can be modified to store energy	Energy is stored in capacitors and superconducting magnets	Conventional capacitors can only store a limited amount of current; they are used as short-term storage devices	Self-discharge rates and costs are high

# Table 2. Details of ESS classification.

# 2.3. Microgrid Loads

Microgrids have several load types vital in their operation, stability, and control. In addition to providing power to several residential, campus, and commercial loads, the microgrid can deliver power to sensitive or critical loads, which demand high reliability. Figure 6 illustrates the load types of microgrids. Several factors should be considered in this scenario, which include prioritizing critical loads, improving the quality and reliability of power for specific loads, and enhancing the reliability of predefined loads. Furthermore, local generation serves as a proactive measure against unforeseen disruptions, supported by swift and precise protection systems [37,61].



Figure 6. Microgrid load types.

# 3. Campus Microgrid Overview

The campus consumes considerable electricity, making relying entirely on the utility grid inconvenient. Therefore, hybrid energy sources are an optimal solution to reduce the cost of campus operation. As a result, several universities have installed microgrids, including photovoltaic (PV), wind turbine (WT), geothermal energy (GE), combined heat and power (CHP), and diesel generators (DGs). Installing renewable energy sources depends on factors such as meteorology and campus size. Therefore, the following campuses have installed varying renewable and conventional resources.

Oakland University (USA) installed hybrid geothermal, PV, solar thermal, and CHP energies [62–66]. Researchers in [67] developed an optimal hybrid renewable energy system for microgrid applications, integrating PV, ESS, and WT, and analyzed its performance and efficiency using HOMER software (Version 3.14.5). Figure 7 shows a microgrid at Oakland University. The study uncovered the potential for economical and eco-friendly energy strategies.



Figure 7. Oakland University microgrid [68].

The Illinois Institute of Technology (USA) utilized distributed generation (DG), controllable loads, storage, and switches. In this study, a high-reliability distribution system (HRDS) optimization technique was applied to achieve significant results in terms of operational cost efficiency. The annual operational costs were significantly reduced from USD 140,497 to USD 126,644 per year by 9% [69]. This demonstrates how the energy management system can dramatically reduce operational costs while increasing energy efficiency. Figure 8 shows the Illinois Institute of Technology's microgrid.



Figure 8. Microgrid at the Illinois Institute of Technology [70].

A comprehensive energy infrastructure has been established at Genoa University (Italy), incorporating diverse components such as a PV plant, three solar thermodynamics dishes, three cogenerating micro-turbines, two natural gas boilers, a refrigerating and absorbing plant, and two electrochemical/thermal storages. This intricate system provides a variety of electrical loads, including the integration of charging units for electric vehicles (EVs), seamlessly interfaced with the E-car Operation Center platform, facilitating both Grid-to-Vehicle (G2V) and Vehicle-to-Grid (V2G) functionalities. Figure 9 illustrates solar panels and a micro-turbine at the Savona Campus. Furthermore, within the confines of the campus, a district heating system is deployed to fulfill heating demands. This system is augmented by the thermal output of micro-turbines and boilers, all operating under the oversight of an Energy Management System (EMS) developed by the University of Genoa [71].



Figure 9. Solar panels and a micro-turbine at Savona Campus, Genoa University [71].

The University of Coimbra (Portugal) has established a progressive energy system featuring PV plants, lithium-ion batteries, EVs, and advanced controllers. The system's performance and efficiency were analyzed using LabVIEW software. This analysis revealed that the university's energy system significantly lowered energy consumption and successfully met 22.3% of the campus's annual electricity demand. Figure 10 shows the microgrid at the University of Coimbra. Therefore, the system proves its dedication to sustainable energy and its efficiency in fulfilling a significant portion of its energy requirements. The annual energy generation is 115.6 MWh/year, while the specific energy yields 1466 MWh annually. The ratio between the PV system's theoretical and actual energy outputs is 88.2%. The energy sent to the grid is 6.5 MWh, whereas the energy consumed internally is 109 MWh [72].



Figure 10. Microgrid at the University of Coimbra [73].

The University of Connecticut (USA) has developed an advanced microgrid system, incorporating various sustainable energy sources. This system includes PV, WT, fuel cell panels, and hydro-kinetic systems, including an ESS. In addition, the natural gas fuel cell located outside the facility energy center supports the campus's microgrid. The efficiency and viability of these components were analyzed using HOMER software. Figure 11 demonstrates the microgrid at the University of Connecticut. The final configuration of the selected microgrid at the University comprises a significant solar-PV capacity of 203,327 kW, a WT system rated at 225,000 kW, and a robust ESS with a capacity of 730,968 kWh. This setup exemplifies the University of Connecticut's commitment to leveraging diverse and renewable energy sources for sustainable and efficient energy management within its campus infrastructure [74].



Figure 11. Microgrid at the University of Connecticut [75].

The Islamic University of Madinah (KSA) proposed an optimal microgrid system combining solar PV, wind energy, and a hybrid alternative. The system's performance and efficiency were analyzed using HOMER software. The proposed microgrid system aims to integrate renewable energy efficiently within Saudi universities. The PV system could cover 3.03% of the university's annual electricity needs with a payback period of 18.6 years. Although the wind system had a higher capacity factor, it had a more extended payback period due to higher costs and less favorable wind conditions. Figure 12 shows the Islamic University of Madinah's microgrid. The hybrid system provided a balanced solution with a 3.7% renewable fraction and a 20.7-year payback period. Both PV and WT systems significantly reduced CO<sub>2</sub>, SO<sub>2</sub>, and NOx emissions, aligning with the university's sustainability goals and Saudi Arabia's broader energy strategies [11].



Figure 12. Islamic University of Madinah's microgrid [76].

The University of California, San Diego (UAS) has installed a microgrid system that provides electrical, heating, and cooling services for a 450-hectare campus accommodating a daily population of 45,000. Comprising two 13.5 MW gas turbines, a 3 MW steam turbine, and a 1.2 MW solar-cell array, this system collectively caters to 85% of the campus's electricity demand and 95% coverage for heating and cooling needs. Noteworthy is the environmental efficiency of the turbines, emitting 75% fewer criteria pollutants than a standard gas power plant. The heating, ventilation, and air conditioning (HVAC) system incorporates a 140,674 kW/h thermal energy storage bank with a capacity of 14,385 m<sup>3</sup>, complemented by three steam turbine-driven chillers and five electricity-driven chillers. Additionally, California's self-generation sponsored a 2.8 MW molten carbonate fuel cell utilizing waste methane. Figure 13 shows solar panels in the microgrid at the University of California, San Diego. The campus connects to San Diego Gas and Electric (SDG&E) through a single 69 kV substation, employing a straight SCADA system for seamless communication between building systems and energy supply. UCSD is currently integrating an advanced master controller (Paladin) to oversee generation, storage, and loads. Operated with hourly computing for optimal conditions, Paladin can process up to 260,000 data inputs per second. Supporting Paladin is the VPower software, which analyzes market-price signals, weather forecasts, and resource availability. Monitoring is facilitated by approximately 200 power meters on main lines and at building main circuit breakers, tracking usage on a minute-by-minute basis [77].



Figure 13. Solar panels in the microgrid at the University of California, San Diego [77].

Nnamdi Azikiwe University (Nigeria) implemented a hybrid energy system, combining solar PV panels and a DG. Figure 14 illustrates the Nnamdi Azikiwe University microgrid. The system's performance and cost-effectiveness were assessed using HOMER analysis. This evaluation determined that the project's Net Present Value (NPV) was USD 1,738,994, and the Levelized Cost of Energy (LCOE) was calculated to be USD 0.264. This analysis indicates that the university's effort to integrate renewable energy solutions alongside traditional power sources aims for a more sustainable and economically viable energy infrastructure [78].



Figure 14. Nnamdi Azikiwe University's microgrid [78].

The primary constituent of the microgrid at Princeton University (USA) installed is a 15 MW gas turbine, which is augmented by 4.5 MW of PV power. During Hurricane Sandy, Princeton's gas-powered CHP facility supplied electricity, heating, and ventilation, ensuring the university's operations continued despite the widespread darkness that engulfed most of the state. The microgrid at Princeton is typically linked to the local grid for operation [79]. Figure 15 illustrates Princeton University's CHP plant microgrid.



Figure 15. Princeton University's CHP plant microgrid [80].

Griffith University's Nathan Campus (Australia) has effectively implemented an advanced energy management system. This system integrates distributed generation (DG) and an ESS with a battery bank, 1164 solar panels, TWs, and full cells (FCs). Figure 16 shows the microgrid at Griffith University's Nathan Campus. An essential feature of this configuration emphasizes effectively controlling and managing a variety of energy sources to accommodate alternating current (AC) and DC loads, including those for EVs. This approach aims at energy management and underscores the university's dedication to sustainable energy practices and its role in pioneering solutions for contemporary energy challenges [81].



Figure 16. Microgrid at Griffith University's Nathan Campus [82].

Nanyang Technological University (Singapore) has implemented a cutting-edge Microgrid Energy Management System (MG-EMS). This system comprises PV panels, FC, and natural gas-operated micro-turbines (MTs), all integrated under the Laboratory of Clean Energy Research (LaCER) [83]. Figure 17 illustrates the microgrid at Nanyang Technological University. This system's focus extends to buildings and transportation within the campus, showcasing NTU's commitment to sustainable energy practices. By incorporating various clean energy sources and advanced management systems, Nanyang Technological University stands at the forefront of energy research and application, particularly in academic institutions.



Figure 17. Microgrid at Nanyang Technological University [84].

Table 3 shows microgrid systems at different universities, including resource, solver, and methodology optimizations.

			,	1 0		
Ref.	Campus Name	Resources	Solver/Methodology/ Optimization	Load Types	Contribution	Results
[72]	University of Coimbra, Portugal	PV, ESS	Control algorithms, IoT	HVAC loads	Improved building microgrid flexibility	Increased energy efficiency
[85]	National University of Sciences and Technology, Pakistan	PV plant, ESS, EVs, DG	MILP, ant colony optimization, LP	Campus load	Reduction of operational cost, analysis of DGs, and optimally scheduled ESS	ESS minimizes operational costs from USD 798,560 to USD 756.385
[86]	Guangdong University of Technology, China	-	Self-crossover genetic algorithm, DSM optimization model	Controllable and non-controllable loads, micro-market operations	DSM scheme for microgrids with sub-decision makers	Reduction in electricity cost
[87]	Cochin University of Science and Technology, India	PV, WT, biomass, etc.	Static and time domain simulations, eigenvalue analysis	-	Microgrid setup with renewable energy resources. The small signal stability assessed by eigenvalue analysis confirmed the system's stable operation for a load increment of 1.26 p.u in grid mode and 1.25 p.u in off-grid mode without violating system constraints	RESs meet a major part of power demand with minimal loss. Stable operation confirmed for significant load increments in both grid mode and off-grid mode

 Table 3. An overview of university campuses' microgrids worldwide.

Energies 2024, 17, 2425

Ref.	Campus Name	Resources	Solver/Methodology/ Optimization	Load Types	Contribution	Results
[88]	NFC Institute of Engineering and Technology, Pakistan	PV, ESS, and Evs	LP	Campus load	Integration of PV system, ESS, and EV in a university campus, optimal Energy Management System (EMS)	EMS decreases energy consumption cost by nearly 45%, EV as a source reduces energy cost by 45.58%, EV as a load reduces energy cost by 19.33%, continuous power supply impact analyzed
[89]	University in Southern Java Island, Indonesia	PV power generation plant	HOMER Pro software, feasibility analysis	Campus load	Feasibility analysis of solar energy system, techno-economic analysis, potential contributions, and applicability	Simulation studies for identifying cost-effective configurations
[90]	University Campus in Brazil	PV and BESS	Simulated annealing algorithm	Campus load	EMS coordination, optimal operation of battery system, reduction in energy consumption costs	Minimize campus energy consumption and costs
[91]	Clemson University- Main Campus, UAS	PV and BESS	Emulated virtual inertia, coordination controller	Campus load	Design and operation of a microgrid, seamless transition between grid-connected and islanded modes, IEEE Std 1547.4 (Hybrid Microgrid Controller Analysis and Design for a Campus Grid. DOI: 10.1109/ PEDG.2019.8807566) compliance	Emulation of virtual inertia for resiliency
[92]	Faculty of Technical Sciences in Novi Sad, Serbia	PV, WT, EV, BESS, biogas micro-turbine	Microcontroller, interface, consumers	-	Application of distributed energy resources, technical specifications for stable island mode operation, techno-economic and environmental analysis	Proposal for a microgrid, analysis of technical specifications for stable operation, focus on annual energy production and investment costs, and avoided CO <sub>2</sub> emissions
[93]	U.E.T, Taxila, Pakistan	Solar PV panels, diesel Generator, energy storage system (ESS)	MILP, MATLAB simulations	Campus load	Reduction of operational cost, increased self-consumption from green DGs, reduction in grid electricity cost	Proposed EMS model for institutional microgrid, reductions in grid electricity cost

# Table 3. Cont.

Energies 2024, 17, 2425

		Table 3. Con	t.			
Ref.	Campus Name	Resources	Solver/Methodology/ Optimization	Load Types	Contribution	Results
[67]	Oakland University, USA	Solar PV, ESS, CHP, and WT	HOMER Pro software	Campus load	Optimal planning and design of hybrid renewable energy systems, scalable and flexible MG configurations	Minimization of NPC and LCOE, comprehensive guide for planning and implementing hybrid renewable energy solutions, potential for cost-effective and sustainable energy, addressing unmet load in MG design
[94]	North China Electric Power University, Beijing, China	PV, WT, CHP, ESS, and EVs	MTPSO	Campus load	Optimal scheduling of power sources Consideration of demand response	The model produces favorable outcomes For hybrid energy microgrids. It exhibits superior global search capabilities when compared to PSO. Simulation analysis confirms the model's effectiveness
[95]	University of California, San Diego, USA	PV and full-cell	Economic optimization	Campus load	State-of-the-art microgrid development, 42 MW microgrid, 92% self-generation of annual electricity load	Achieving savings of USD 800,000 per month through microgrid PV panels, improving the existing grid infrastructure, and attaining a high level of self-generation for electricity load
[79]	Princeton University, UAS	Gas turbine (15 MW), solar field (4.5 MW), CHP, ESS	Digital controls	Campus load	Multiple fuel sources, multiple power-generating assets, CHP production, modern digital controls, real-time awareness of fuel and electricity costs	Resilience during Hurricane Sandy, continuity of critical research projects and computing services, lower carbon footprint, higher reliability with behind-the-meter CHP, economic dispatch, underground power distribution, revenue generation through power exports and ancillary services, lessons for successful microgrid operation

Ref.	Campus Name	Resources	Solver/Methodology/ Optimization	Load Types	Contribution	Results
[77]	University of California, San Diego, USA	Two 13.5 MW gas turbines, 3 MW steam turbine, 1.2 MW solar-cell array, 2.8 MW molten carbonate FC, 140,674 kW/h thermal energy storage bank, Paladin master controller, VPower software	Environmental efficiency, Paladin integration, VPower analysis, thermal energy storage, SCADA system	Campus load	85% of electricity, 95% of heating and cooling needs, 75% fewer criteria pollutants, 30% federal investment tax credit	Improved environmental efficiency, significant coverage of campus energy demands, financial incentives for sustainability
[96]	University of Genoa, Italy		Model Productive Control (MPC)	Campus load	Increased overall energy efficiency, lower primary energy consumption, environmental and economic sustainability	Reducing emissions, primary energy use, and costs
[97]	Chiang Mai Rajabhat University, Thailand	PV, DG, and biomass gasifier	-	Smart community	Hybrid PV-DC microgrid system design and evaluation, application of DC loads	An enhanced and practical hybrid PV-DC microgrid system was developed, comprising components such as a PV-AC microgrid, diesel generator, biomass gasifier, and connection to the local grid
[98]	Hangzhou Dianzi University, China	PV, DG, fuel cells, and BESS unit	-	Campus load	The primary power source consists of PV panels supported by a small diesel generator and fuel cells. These are integrated with a capacitor bank and storage battery unit	The world's first microgrid to achieve a 50% PV penetration rate utilized 728 solar panels covering 946 m <sup>2</sup> . It established a stable microgrid system despite the high penetration of intermittent

Table 3. Cont.

Table 3. Cont.

Ref.	Campus Name	Resources	Solver/Methodology/ Optimization	Load Types	Contribution	Results
[99]	Technical University of Denmark	PV, WT, and vanadium-based battery system	Experimental tests for static and dynamic stability analysis	Laboratory scale load	The implementation and testing of various control strategies for the combined system followed the development of appropriate models for the SYSLAB microgrid	Test findings encompassed static and dynamic stability, the impact of disturbances on power system equipment and network, parameters for dynamic modeling of DER components, and the creation of appropriate models for the SYSLAB microgrid
[100]	Nigerian University	PV panels, inverter, grid system, DG set	HOMER Pro software	University community load	A microgrid system was designed and sized to tackle power challenges within the Nigerian national grid	PV panels have the potential to generate 88.0% of the campus's annual energy, leading to an 88.0% reduction in the university community's electricity bill and CO <sub>2</sub> emissions

# 4. The Proposed Optimization Techniques

Numerous different optimization techniques have been proposed to analyze the campus microgrid performance systems. Among these techniques, Genetic Algorithm (GA), Particle Swarm Optimization (PSO), Tuna Swarm Optimization (TSO), Cuckoo Search (CS), Grey Wolf Optimizer (GWO), and Gradient-based Grey Wolf Optimizer (GGWO), the Hybrid Optimization of Multiple Electric Renewables (HOMER), Firefly Algorithm (FA), LabVIEW Simulation Model (LSM), Mixed Integer Linear Programming (MILP) [101], nonlinear programming [90], High-Reliability Distribution System (HRDS), YALMIP toolbox of MATLAB, Mixed Integer Conic Programming (MICP), and Quantum Teaching Learning-Based Optimization (QTLBO), NSGA-II, and EDNSGA-II [102–107]. Sardou et al. [108] proposed a robust algorithm that integrates the PSO algorithm with the primal-dual interior point (PDIP) method for the efficient management of microgrid energy. Jaramillo et al. [109] developed the MILP algorithm to optimize microgrid operation. Guo et al. [110] designed an economically optimal energy management model that combines the dynamic programming technique with the grid input and output strategy for grid-connected PV systems. Optimizing the size of the components for an islanded hybrid PV/WT system with an integrated energy management system was carried out by Rullo, P. et al. [111]. The optimization approach was based on an economic model predictive control (EMPC) model. Furthermore, Indragandhi, V. et al. [112] proposed a Multi-Objective Particle Swarm Optimization (MOPSO) algorithm to investigate AC/DC microgrid power management. Mellouk et al. [113] developed the Parallel Genetic-Particle Swarm Optimization Algorithm (PGPSO) to address the challenges of solving energy management optimization problems and determining the appropriate size of renewable energy components. Li et al. [106]

introduced a novel approach that combines incremental conductance (INC) and Improved Tuna Swarm Optimization Hybrid INC (ITSO-INC) to accurately track the maximum power point. Moreover, Rajagopalan et al. [107] enhanced the Oppositional Gradient-based Grey Wolf Optimizer (OGGWO) algorithm to clarify the microgrids' optimal operation.

Authors in [114] proposed a novel optimization technique for grid-connected solar PV to minimize the total life cycle cost and energy purchased from the utility grid while maximizing reliability, considering the instability of solar energy. The mixed integer linear programming study focuses on the loss of power supply probability (LPSP) to measure microgrid system reliability.

However, the mathematical models include the Genetic Algorithm (GA), Genetic Programming (GP), Differential Evolution (DE), Artificial Bee Colony (ABC), Simulated Annealing (SA), and Particle Swarm Optimization (PSO) [115,116]. Some optimization techniques perform better in terms of effectiveness and accuracy. For instance, Güven, A. F. et al. [117] compared GA, PSO, FA, HOMER, and a novel Firefly and PSO algorithm (HFAPSO) hybrid to ensure the hybrid microgrid's optimal sizing. The results revealed that the HFAPSO was the most effective algorithm compared to the techniques mentioned. Hertzog, P. E et al. [118] concluded that the LabVIEW simulation model (LSM) was more confident than HOMER. Furthermore, Li et al. [106] proved that the ITSO-INC algorithm outperformed both the CS and TSO algorithms in high accuracy, fast response to dynamic changes, rapid convergence, and the lack of steady-state oscillation. Rajagopalan et al. [93] found that the OGGWO algorithm surpassed the PSO, CS, GWO, GGWO, NSGA-II, and EDNSGA-II to reduce costs and mitigate pollution.

#### 5. An Overview of Campus Microgrid Architectures

Microgrid hybrid systems typically consist of four components: photovoltaics (PVs), energy storage systems (ESSs), wind turbines (WTs), and combined heat and power (CHP). The configuration of the microgrid system depends upon considering factors such as campus size, climatic conditions, and geographical location. These factors have a substantial impact on the overall performance of the system. The microgrid components are organized into various configurations, which are categorized as follows:

## 5.1. Configuration A: (Hybrid PV-Grid-Connected)

This configuration consists of a utility grid, PV panel, converter, inverter, and energy storage, as shown in Figure 18. PV panels are regarded as the primary renewable energy source that harnesses sun irradiation to generate power. Subsequently, the harvested power obtained from PV panels in direct current (DC) requires an inverter to convert it into alternating current (AC) for compatibility with power-consumption devices. Diesel generators (DGs) are used during power outages to provide electricity to individual buildings on campus.

The advantages of a hybrid PV-grid connection are that it is simple, free to operate, and sometimes does not require energy storage, such as when solar power generates less electricity than the campus demands. However, on the other hand, PV panels are influenced by weather conditions, which could limit their ability to generate power. Moreover, the hybrid PV-grid-connected system is commonly installed in residential and commercial buildings due to its simplicity. For instance, many university campuses worldwide have installed hybrid PV-grid-connected systems, such as the University of Energy and Natural Resources [119], GITAM Deemed To be University in Andhra Pradesh [120], University Malaysia Pahang [121], Effat University [122], Heriot-Watt University [123], Chiba University of Commerce [124], The Hashemite University [125], University at Albany [126], Aga Khan University [127], Colorado State University [128], University of Northern Colorado [129], University of Wisconsin [130], Iowa State University [131], University of Nottingham, Bath University, Exeter University and University of the West of England [132], and Washington and Lee University [133].



Figure 18. Grid connected with solar PV and energy storage.

It is important to note that the required energy storage system (ESS) depends on the amount of solar capacity installed on the campus. When the power produced by the PV panels exceeds the consumption demand, an energy storage system is required. For example, the University of Texas [134], British Malaysian Institute [135], Mulhouse campus [136], North Central College [137], Gavilan College [138], Sanford Burnham Prebys Campus [139], New Mexico State University [140], and California State University [141]. Other university campuses such as Beloit College, Fairleigh Dickenson University, Georgia Tech, Lake Superior College, Lane Community College, Luther College, Northern Arizona University, Milwaukee Area Technical College, South Central College, Thomas College, Tuskegee University of Minnesota Duluth, and Washington and Lee University are suitable for installing solar power and energy storage systems owing to their weather conditions and geographical locations according to the National Renewable Energy Laboratory [142].

## 5.2. Configuration B (Hybrid PV-WT-Grid-Connected)

This hybrid configuration uses photovoltaic panels, wind turbines, a utility grid, converters, and inverters, among other essential components. Figure 19 illustrates the components of this configuration; WT and PV panels are considered the main sources of renewable energy. The National Renewable Energy Laboratory (NREL) states that wind turbines can be installed in distributed wind (that is, small turbines on site), land-based wind (i.e., farms and forests), and offshore wind (i.e., ocean) [143]. Since wind generators work best in an isolated environment, few universities have installed wind turbines, such as Quinnipiac University [144], University of Delaware [145], Carleton College, University of Minnesota [146], St. Olaf College [147], and Macalester College, St. Paul [148].



Figure 19. Microgrid connected with solar PV, WT, and ESS.

# 5.3. Configuration C (Hybrid PV-Diesel Generator-Grid Connected)

Like the one shown in Figure 20, a hybrid system consists of PV, DG, and grid connection. This configuration has been installed at the following universities: Florida International University [149], University of the Free State [150], University of KwaZulu Natal [2], and Jomo Kenyatta University of Agriculture and Technology [151].



Figure 20. Grid connected with PV, diesel generator, and energy storage.

# 5.4. Configuration D (Hybrid PV-CHP-Grid Connected)

This configuration illustrates the usage of a hybrid system composed of PV, CHP, utility grid, converter, and an inverter. The CHP allows electricity and heat to be produced simultaneously, compared with generating them separately [152]. Microgrid generation and consumption are characterized by distinctive properties, which increase their flexibility when utilizing renewable energy and CHP [153]. Thus, centralized grids are more adaptable. Figure 21 illustrates the components of this configuration.

21 of 29



Figure 21. Microgrid connected with PV, CHP, and ESS.

Other universities have installed this configuration, such as Oakland University [54], University of the West of England [140], Kent State University [141], Chalmers University of Technology [142], Stanford University [143], Clemson University [144], IST—Alameda Campus [145], Rowan University [146], the University Campus of UNICAMP [147], and the University of Genoa [154].

Numerous universities worldwide have installed microgrids on their campuses, as shown in Figure 22. The installations of configuration A, configuration B, configuration C, and configuration D are %0.54, %0.08, %0.07, and %0.14, respectively. The most common configuration used is configuration A due to its simplicity and free-charge operation, which does not require energy storage and is cheaper than other configurations. However, configuration C is less common due to the high cost of fossil fuels and the caused air pollution.



# Campus microgrid architecture

**Figure 22.** Comparison of the configurations, where configuration A is (Hybrid PV-grid-connected), configuration B is (Hybrid PV-WT-grid-connected), configuration C is (Hybrid PV-diesel generator-grid-connected), and configuration D is (Hybrid PV-CHP-grid-connected).

As a result, this study comprehensively reviews microgrid systems on university campuses, covering principles, types, graphical locations, used algorithms, connections, and applications. It also undertakes a comparative analysis of geothermal (GE), wind turbines (WTs), and photovoltaics (PVs) in terms of installation cost, availability, weather conditions, efficiency, environmental impact, and maintenance. Furthermore, the paper introduces and compares hybrid microgrid system configurations, aiming to identify the optimal configuration for energy production and flexibility.

# 6. Conclusions

This paper comprehensively reviewed the pending university campus microgrids regarding principles, types, geographical locations, algorithms, connections, and applications. Some renewable energy sources, such as geothermal (GE), wind turbine (WT), and photovoltaic (PV), were compared in terms of installation costs, availability, weather conditions, efficiency, environmental impact, and maintenance (see Table 1). Furthermore, a description of microgrid systems and their components, including distributed generation (DG), energy storage system (ESS), and microgrid load was presented. The most common optimization models for analyzing the performance of campus microgrids were discussed. Some optimization techniques perform better in terms of effectiveness and accuracy. For instance, novel Hybrid Firefly Algorithms and Particle Swarm Optimization (HFAPSO), Improved Tuna Swarm Optimization and Incremental Conductance (ITSO-INC) algorithm, and Oppositional Gradient-based Grey Wolf Optimizer (OGGWO) algorithm. Also, the LabVIEW Simulation Model (LSM) was more confident than HOMER. Hybrid microgrid system configurations were introduced and compared to find the optimal configuration in terms of energy production and flexibility. Hence, configuration A (Hybrid PV-gridconnected) was the most common configuration used compared to the others (see Figure 22) owing to its simplicity and free-charge operation, which does not require energy storage and is cheaper than other configurations. However, configuration C is less common due to the high cost of fossil fuels and the causes of air pollution.

**Author Contributions:** Conceptualization, E.Y.A.; methodology, E.Y.A.; validation, E.Y.A. and K.S.; formal analysis, K.S.; resources, E.Y.A.; data curation, E.Y.A.; writing—original draft preparation, E.Y.A.; writing—review and editing, E.Y.A. and K.S.; visualization, E.Y.A.; supervision, M.A.Z. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Conflicts of Interest: The authors declare no conflict of interest.

# Abbreviations

ABC	Artificial bee colony
AC	Alternating current
BESS	Battery energy storage system
CHP	Combined heat and power
CO <sub>2</sub>	Carbon dioxide
CS	Cuckoo search
DC	Direct current
DE	Differential evolution
DG	Diesel generator
DG	Distributed generation
DSM	Demand side management
EDNSGA-II	Economic dispatch-based non-dominated sorting genetic algorithm II
EMPC	Economic model predictive control
EMS	Energy management system
ESS	Energy storage system
EVs	Electric vehicles
FA	Firefly algorithm

FC	Full cell
GA	Genetic algorithm
GA	Genetic programming
GE	Geothermal energy
GGWO	Gradient-based grey wolf optimizer
GW	Gigawatts
GWh	Gigawatt-hour
GWO	Grey wolf optimizer
HFAPSO	Hybrid of FA and PSO algorithms
HOMER	Hybrid optimization of multiple energy resources
HPS	Hybrid power systems
HRDS	High-reliability distribution system
IEA	International energy agency
INC	Incremental conductance
IoT	Internet of things
KW	Kilowatts
LCOE	Levelized cost of energy
LP	Linear programming
LSM	LabVIEW simulation model
MICP	Mixed-integer conic programming
MILP	Mix integer linear programming
MOPSO	Multi-objective particle swarm optimization
MTPSO	Multi-team particle swarm optimization
MW	Megawatts
NPC	Net present cost
NSGA-II	Non-dominated sorting genetic algorithm II
PDIP	Primal-dual interior point
PGPSO	Parallel genetic-particle swarm optimization algorithm
PSO	Particle swarm optimization
PV	Photovoltaic
QTLBO	Quantum teaching learning-based optimization
RES	Renewable energy sources
SCADA	Supervisory control and data acquisition
TSO	Tuna swarm optimization
ΤW	Terawatts
WT	Wind turbine

# References

- 1. Salhein, K.; Kobus, C.J.; Zohdy, M. Forecasting Installation Capacity for the Top 10 Countries Utilizing Geothermal Energy by 2030. *Thermo* **2022**, *2*, 23. [CrossRef]
- 2. Akindeji, K.T.; Ewim, D.R.E. Economic and environmental analysis of a grid-connected hybrid power system for a University Campus. *Bull. Natl. Res. Cent.* **2023**, *47*, 75. [CrossRef]
- 3. Hyder, Z. Geothermal Energy Pros and Cons. 2023. Available online: https://www.solarreviews.com/blog/geothermal-energy-pros-and-cons (accessed on 25 January 2024).
- 4. Bayindir, R.; Hossain, E.; Kabalci, E.; Perez, R. A comprehensive study on microgrid technology. *Int. J. Renew. Energy Res.* 2014, *4*, 1094–1107.
- 5. Mariam, L.; Basu, M.; Conlon, M.F. A Review of Existing Microgrid Architectures. J. Eng. 2013, 2013, 937614. [CrossRef]
- 6. Marsh, J. Geothermal vs. Solar Energy: A Comparative Analysis. 2023. Available online: https://photovoltaicsolarenergy.org/geothermal-vs-solar-energy-a-comparative-analysis/ (accessed on 27 January 2024).
- 7. Geographic, N. Geothermal Energy. Available online: https://education.nationalgeographic.org/resource/geothermal-energy/ (accessed on 15 January 2024).
- Fernández, L. Weighted Average Cost for Installed Onshore Wind Energy Worldwide from 2010 to 2022. 2023. Available online: https://www.statista.com/statistics/506774/weighted-average-installed-cost-for-onshore-wind-power-worldwide/#: ~:text=Global%20average%20cost%20for%20installed%20onshore%20wind%20projects%202010-2022&text=In%202022,%20 the%20cost%20for,comparison%20to%20the%20previous%20year (accessed on 25 January 2024).
- Parkman, K. How Much Do Solar Panels Cost in 2024? 2024. Available online: https://www.consumeraffairs.com/solar-energy/ how-much-do-solar-panels-cost.html#:~:text=Solar%20panels%20generate%20%E2%80%9Cfree%E2%80%9D%20electricity, %20but%20installing%20a,factoring%20in%20the%2030%25%20federal%20solar%20tax%20credit (accessed on 25 January 2024).

- 10. Fernández, L. Average Installed Cost for Geothermal Energy Worldwide from 2010 to 2022. 2023. Available online: https://www.statista.com/statistics/1027751/global-geothermal-power-installation-cost-per-kilowatt/ (accessed on 25 January 2024).
- 11. International Renewable Energy Agency. Solar Energy 2023. Available online: https://www.irena.org/Energy-Transition/ Technology/Solar-energy (accessed on 25 January 2024).
- 12. Geothermal Energy. 2018. Available online: https://www.encyclopedia.com/science-and-technology/technology/technology-terms-and-concepts/geothermal-energy2018 (accessed on 25 January 2024).
- 13. National Renewable Energy Laboratory and the Department of Energy. Geothermal Electricity Production. Available online: https://www.renewableenergyworld.com/types-of-renewable-energy/tech-3/geoelectricity/#gref (accessed on 25 January 2024).
- Klise, G.T.; Hill, R.; Walker, A.; Dobos, A.; Freeman, J. PV system "Availability" as a reliability metric—Improving standards, contract language and performance models. In Proceedings of the 2016 IEEE 43rd Photovoltaic Specialists Conference (PVSC), Portland, OR, USA, 5–10 June 2016; IEEE: Piscataway, NJ, USA, 2016.
- 15. Gi, O.; Adegoke, C.W. Analysis of Solar Radiation Availability for Deployment of Solar Photovoltaic (PV) Technology in a Tropical City. *Int. J. Appl.* **2015**, *4*, 41–46.
- 16. Einarsson, S. Wind Turbine Reliability Modeling. Ph.D. Thesis, Reykjavik University, Reykjavik, Iceland, 2016.
- 17. Kim, K.-H.; Kim, J.-Y.; Hwang, S.-J.; Yoon, C.-Y.; Kim, H.-G. Research on Suitable Array Power Installation Density of Wind Turbines for the Assessment of Wind Energy Potential for Data Platform. *J. Korean Sol. Energy Soc.* **2023**, *43*, 13–23. [CrossRef]
- 18. Pedraza, J.M. Chapter 5—The Use of Geothermal Energy for Electricity Generation. 2022. Available online: https://www.sciencedirect.com/science/article/abs/pii/B978012823440200020?via=ihub (accessed on 3 May 2024).
- 19. Kurpaska, S.; Knaga, J.; Latała, H.; Sikora, J.; Tomczyk, W. Efficiency of solar radiation conversion in photovoltaic panels. *BIO Web Conf.* **2018**, *10*, 02014. [CrossRef]
- Salim; Ohri, J. Performance Study of LabVIEW Modelled PV Panel and Its Hardware Implementation. Wirel. Pers. Commun. 2022, 123, 2759–2774. [CrossRef]
- 21. Blackwood, M. Maximum efficiency of a wind turbine. Undergrad. J. Math. Model. One + Two 2016, 6, 2. [CrossRef]
- 22. Pellegri, A. The complementary betz theory. *arXiv* 2022, arXiv:2201.00181.
- 23. Huang, J.; Abed, A.M.; Eldin, S.M.; Aryanfar, Y.; Alcaraz, J.L.G. Exergy analyses and optimization of a single flash geothermal power plant combined with a trans-critical CO<sub>2</sub> cycle using genetic algorithm and Nelder–Mead simplex method. *Geotherm. Energy* **2023**, *11*, 4. [CrossRef]
- 24. Dale, M. A Comparative Analysis of Energy Costs of Photovoltaic, Solar Thermal, and Wind Electricity Generation Technologies. *Appl. Sci.* **2013**, *3*, 325–337. [CrossRef]
- Gómez, J.C.L.; Aldaco, S.E.D.L.; Alquicira, J.A. A Review of Hybrid Renewable Energy Systems: Architectures, Battery Systems, and Optimization Techniques. *Eng* 2023, 4, 84. [CrossRef]
- Chen, H.; Cong, T.N.; Yang, W.; Tan, C.; Li, Y.; Ding, Y. Progress in electrical energy storage system: A critical review. *Prog. Nat. Sci.* 2009, 19, 291–312. [CrossRef]
- Abu-Sharkh, S.; Arnold, R.J.; Kohler, J.; Li, R.; Markvart, T.; Ross, J.N.; Yao, R. Can microgrids make a major contribution to UK energy supply? *Renew. Sustain. Energy Rev.* 2006, 10, 78–127. [CrossRef]
- Basu, M. Economic environmental dispatch using multi-objective differential evolution. *Appl. Soft Comput.* 2011, 11, 2845–2853. [CrossRef]
- 29. Kok, N.; Miller, N.; Morris, P. The economics of green retrofits. J. Sustain. Real Estate 2012, 4, 4–22. [CrossRef]
- Valant, C.; Gaustad, G.; Nenadic, N. Characterizing Large-Scale, Electric-Vehicle Lithium Ion Transportation Batteries for Secondary Uses in Grid Applications. *Batteries* 2019, 5, 8. [CrossRef]
- 31. Dohn, R.L. The business case for microgrids. White Pap. Siemens 2011, 6-8.
- 32. Marnay, C. Microgrids and heterogeneous security, quality, reliability, and availability. In Proceedings of the 2007 Power Conversion Conference-Nagoya, Nagoya, Japan, 2–5 April 2007; IEEE: Piscataway, NJ, USA, 2007.
- Al Dakhl, S.; Alali, D.; Habbi, H.M.; Zohdy, M. Security Issues on Smart Grid and Blockchain-Based Secure Smart Energy Management Systems. In Proceedings of the 2023 IEEE 11th International Conference on Smart Energy Grid Engineering (SEGE), Oshawa, ON, Canada, 13–15 August 2023; pp. 194–200.
- Akindeji, K.T.; Tiako, R.; Davidson, I. Optimization of University Campus Microgrid for Cost Reduction: A Case Study. Adv. Eng. Forum 2022, 45, 77–96. [CrossRef]
- 35. Akindeji, K.T. Performance and Cost Benefit Analyses of University Campus Microgrid. Ph.D. Thesis, University of London, London, UK, 2021.
- Hadjidemetriou, L.; Zacharia, L.; Kyriakides, E.; Azzopardi, B.; Azzopardi, S.; Mikalauskiene, R.; Al-Agtash, S.; Al-Hashem, M.; Tsolakis, A.; Ioannidis, D.; et al. Design factors for developing a university campus microgrid. In Proceedings of the 2018 IEEE International Energy Conference (ENERGYCON), Limassol, Cyprus, 3–7 June 2018; IEEE: Piscataway, NJ, USA, 2018; pp. 1–6.
- Lidula, N.; Rajapakse, A. Microgrids research: A review of experimental microgrids and test systems. *Renew. Sustain. Energy Rev.* 2011, 15, 186–202. [CrossRef]
- Fulghum, N. Energy and Climate Data Analyst. 2024. Available online: https://ember-climate.org/data-catalogue/yearlyelectricity-data/ (accessed on 29 January 2024).

- Global Wind Energy Council. Global Wind Energy Outlook 2000 Gigawatts by 2030. Available online: https://gwec.net/globalwind-energy-outlook-2000-gigawatts-2030/#:~:text=Global%20wind%20power%20capacity%20could,Energy%20Council%20 and%20Greenpeace%20International (accessed on 29 January 2024).
- 40. Yuen, S. Global Solar PV Capacity to Increase 1TW Annually by 2030—InfoLink. 2023. Available online: https://www.pv-tech. org/global-solar-pv-capacity-to-increase-1tw-annually-by-2030-infolink/#:~:text=firm%20InfoLink%20Consulting.-,The% 20global%20solar%20PV%20market%20installed%20250GW%20of%20new%20capacity,6TW%20in%20the%20same%20year (accessed on 22 January 2024).
- 41. INTERNATIONAL ENERGY AGENCY. Energy Transitions in G20 Countries. 2018. Available online: https://biblioteca.olade. org/opac-tmpl/Documentos/cg00692.pdf (accessed on 13 January 2024).
- 42. Tercan, S.M.; Demirci, A.; Gokalp, E.; Cali, U. Maximizing self-consumption rates and power quality towards two-stage evaluation for solar energy and shared energy storage empowered microgrids. *J. Energy Storage* **2022**, *51*, 104561. [CrossRef]
- 43. Orioli, A.; Di Gangi, A. Five-years-long effects of the Italian policies for photovoltaics on the energy demand coverage of grid-connected PV systems installed in urban contexts. *Energy* **2016**, *113*, 444–460. [CrossRef]
- 44. KHOLER Company. Energy Storage. Available online: https://www.clarke-energy.com/us/energy-storage/ (accessed on 15 January 2024).
- 45. Linquip Team. 10 Main Types of Energy Storage Methods in 2023. Available online: https://www.linquip.com/blog/types-ofenergy-storage-methods/ (accessed on 20 January 2024).
- 46. Ibrahim, H.; Ilinca, A.; Perron, J. Energy storage systems—Characteristics and comparisons. *Renew. Sustain. Energy Rev.* 2008, 12, 1221–1250. [CrossRef]
- 47. Abah, J.U. Optimisation of Renewable Energy Microgrid Systems for Developing Countries. Ph.D. Thesis, University of Central Lancashire, Preston, UK, 2021.
- 48. Diaz-Gonzalez, F.; Bianchi, F.D.; Sumper, A.; Gomis-Bellmunt, O. Control of a Flywheel Energy Storage System for Power Smoothing in Wind Power Plants. *IEEE Trans. Energy Convers.* **2013**, *29*, 204–214. [CrossRef]
- 49. Palizban, O.; Kauhaniemi, K. Energy storage systems in modern grids—Matrix of technologies and applications. *J. Energy Storage* **2016**, *6*, 248–259. [CrossRef]
- Li, W.; Joos, G. Comparison of Energy Storage System Technologies and Configurations in a Wind Farm. In Proceedings of the 2007 IEEE Power Electronics Specialists Conference, Orlando, FL, USA, 17–21 June 2007; IEEE: Piscataway, NJ, USA, 2007; pp. 1280–1285.
- 51. Hannan, M.; Hoque, M.; Mohamed, A.; Ayob, A. Review of energy storage systems for electric vehicle applications: Issues and challenges. *Renew. Sustain. Energy Rev.* 2017, *69*, 771–789. [CrossRef]
- 52. Alsaidan, I.; Khodaei, A.; Gao, W. A Comprehensive Battery Energy Storage Optimal Sizing Model for Microgrid Applications. *IEEE Trans. Power Syst.* 2017, 33, 3968–3980. [CrossRef]
- 53. Inthamoussou, F.A.; Pegueroles-Queralt, J.; Bianchi, F.D. Control of a Supercapacitor Energy Storage System for Microgrid Applications. *IEEE Trans. Energy Convers.* 2013, 28, 690–697. [CrossRef]
- 54. Leou, R.-C. An economic analysis model for the energy storage system applied to a distribution substation. *Int. J. Electr. Power Energy Syst.* **2012**, *34*, 132–137. [CrossRef]
- 55. Ibrahim, H.; Belmokhtar, K.; Ghandour, M. Investigation of Usage of Compressed Air Energy Storage for Power Generation System Improving—Application in a Microgrid Integrating Wind Energy. *Energy Procedia* **2015**, *73*, 305–316. [CrossRef]
- 56. Subburaj, A.S.; Pushpakaran, B.N.; Bayne, S.B. Overview of grid connected renewable energy based battery projects in USA. *Renew. Sustain. Energy Rev.* **2015**, *45*, 219–234. [CrossRef]
- 57. Ma, T.; Yang, H.; Lu, L. A feasibility study of a stand-alone hybrid solar–wind–battery system for a remote island. *Appl. Energy* **2014**, *121*, 149–158. [CrossRef]
- Lasseter, R.H. Microgrids. In Proceedings of the 2002 IEEE Power Engineering Society Winter Meeting, New York, NY, USA, 27–31 January 2002; Conference proceedings (Cat. No. 02CH37309). IEEE: Piscataway, NJ, USA, 2002.
- 59. Aghamohammadi, M.R.; Abdolahinia, H. A new approach for optimal sizing of battery energy storage system for primary frequency control of islanded Microgrid. *Int. J. Electr. Power Energy Syst.* **2014**, *54*, 325–333. [CrossRef]
- 60. Di Silvestre, M.L.; Graditi, G.; Sanseverino, E.R. A Generalized Framework for Optimal Sizing of Distributed Energy Resources in Micro-Grids Using an Indicator-Based Swarm Approach. *IEEE Trans. Ind. Inform.* **2013**, *10*, 152–162. [CrossRef]
- 61. Planas, E.; Gil-De-Muro, A.; Andreu, J.; Kortabarria, I.; de Alegría, I.M. General aspects, hierarchical controls and droop methods in microgrids: A review. *Renew. Sustain. Energy Rev.* **2013**, *17*, 147–159. [CrossRef]
- 62. Leidel, J. Integrating Large Scale, Innovative Solar Thermal Systems into the Built Environment. 2013. Available online: https://www.oakland.edu/Assets/upload/docs/Energy/Oakland\_University\_ASHRAE\_June\_26\_2013\_web.pdf (accessed on 1 March 2024).
- 63. Salhein, K.; Kobus, C.J.; Zohdy, M. Control of Heat Transfer in a Vertical Ground Heat Exchanger for a Geothermal Heat Pump System. *Energies* **2022**, *15*, 5300. [CrossRef]
- 64. Leidel, J. Human Health Science Building Geothermal Heat Pump Systems. Ph.D. Thesis, Oakland University, Rochester, MI, USA, 2014.
- 65. Salhein, K.A.A. Modeling and Control of Heat Transfer in a Single Vertical Ground Heat Exchanger for a Geothermal Heat Pump System. Ph.D. Thesis, Oakland University, Rochester, MI, USA, 2023.

- 66. Salhein, K.; Kobus, C.J.; Zohdy, M. Heat Transfer Control Mechanism in a Vertical Ground Heat Exchanger: A Novel Approach. *Fundam. Res. Appl. Phys. Sci.* 2023, *5*, 59–90. [CrossRef]
- 67. Alhawsawi, E.Y.; Habbi, H.M.D.; Hawsawi, M.; Zohdy, M.A. Optimal Design and Operation of Hybrid Renewable Energy Systems for Oakland University. *Energies* **2023**, *16*, 5830. [CrossRef]
- 68. Macomb Mechanical. Oakland University—Health and Human Services Building. 2012. Available online: https://www.macombmechanical.com/oakland-university-health-and-human-services-building (accessed on 1 February 2023).
- Shahidehpour, M.; Khodayar, M.; Barati, M. Campus microgrid: High reliability for active distribution systems. In Proceedings of the 2012 IEEE Power and Energy Society General Meeting, San Diego, CA, USA, 22–26 July 2012; IEEE: Piscataway, NJ, USA, 2012; pp. 1–2. [CrossRef]
- 70. Microgrid. Available online: https://www.iit.edu/microgrid (accessed on 15 December 2023).
- The University of Genoa. La Smart Polygeneration Microgrid (SPM). Available online: https://campus-savona.unige.it/en/ progetti/Energia2020/SPM (accessed on 20 December 2023).
- 72. Correia, A.; Ferreira, L.M.; Coimbra, P.; Moura, P.; de Almeida, A.T. Smart Thermostats for a Campus Microgrid: Demand Control and Improving Air Quality. *Energies* **2022**, *15*, 1359. [CrossRef]
- Moura, P.; Correia, A.; Delgado, J.; Fonseca, P.; de Almeida, A. University campus microgrid for supporting sustainable energy systems operation. In Proceedings of the 2020 IEEE/IAS 56th Industrial and Commercial Power Systems Technical Conference (I&CPS), Las Vegas, NE, USA, 29 June–28 July 2020; IEEE: Piscataway, NJ, USA, 2020.
- Kou, W.; Bisson, K.; Park, S.-Y. A Distributed Demand Response Algorithm and its Application to Campus Microgrid. In Proceedings of the 2018 IEEE Electronic Power Grid (eGrid), Charleston, SC, USA, 12–14 November 2018; IEEE: Piscataway, NJ, USA, 2018; pp. 1–6.
- 75. University of Connecticut. Center for Clean Energy Engineering (C2E2). Available online: https://www.energy.uconn.edu/uconn-microgrid/. (accessed on 15 December 2023).
- Islamic University of Madinah. Microgrid at Islamic University of Madinah. Available online: <a href="https://www.arthurerickson.com/unbuilt-projects/islamic-university-of-madinah/3">https://www.arthurerickson.com/unbuilt-projects/islamic-university-of-madinah/3</a> (accessed on 12 December 2023).
- International Microgrid Symposium (ISC). The University of California, San Diego (UCSD). Available online: <a href="https://microgrid-symposiums.org/microgrid-examples-and-demonstrations/uc-san-diego-microgrid/">https://microgrid-symposiums.org/microgrid-examples-and-demonstrations/uc-san-diego-microgrid/</a> (accessed on 10 December 2023).
- Emma Elekwa, O.; Microgrid Nnamdi Azikiwe University. 4 December 2022. Available online: https://thenationonlineng.net/ unizik-biz-school-launches-100000-engineering-grant-project/ (accessed on 10 December 2023).
- University, Case Study: Microgrid at Princeton University. 2015. Available online: https://facilities.princeton.edu/news/casestudy-microgrid-princeton-university#:~:text=An%20example%20is%20the%20microgrid,of%20the%20state%20was%20dark. (accessed on 29 November 2023).
- 80. Borer, T. Case Study: Microgrid at Princeton University. 2015. Available online: https://facilities.princeton.edu/news/casestudy-microgrid-princeton-university (accessed on 28 November 2023).
- Leskarac, D.; Moghimi, M.; Liu, J.; Water, W.; Lu, J.; Stegen, S. Hybrid AC/DC Microgrid testing facility for energy management in commercial buildings. *Energy Build.* 2018, 174, 563–578. [CrossRef]
- 82. Griffith University. Climate Action. 2022. Available online: https://www.griffith.edu.au/sustainability/climate-action (accessed on 19 December 2023).
- 83. Singh, R.; Kumar, K.N.; Sivaneasan, B.; So, P.; Gooi, H.B.; Jadhav, N.; Marnay, C. *Marnay, Sustainable Campus with PEV and Microgrid*; Lawrence Berkeley National Lab. (LBNL): Berkeley, CA, USA, 2021.
- 84. NANYANG TECHNOLOGICAL UNIVERSITY. Institution Rooftop Nanyang Technological University, Singapore. Available online: https://solargy.com.sg/new/image/catalog/NTU.Yingli%20Case%20Study.pdf (accessed on 10 December 2023).
- Hashmi, S.U.H.; Ulasyar, A.; Zad, H.S.; Khattak, A.; Imran, K. Optimal Day-Ahead Scheduling for Campus Microgrid by using MILP Approach. *Pak. J. Eng. Technol.* 2021, *4*, 21–27. [CrossRef]
- Xu, F.; Wu, W.; Zhao, F.; Zhou, Y.; Wang, Y.; Wu, R.; Zhang, T.; Wen, Y.; Fan, Y.; Jiang, S. A micro-market module design for university demand-side management using self-crossover genetic algorithms. *Appl. Energy* 2019, 252, 113456. [CrossRef]
- Saritha, K.S.; Sreedharan, S.; Nair, U. A generalized setup of a campus microgrid—A case study. In Proceedings of the 2017 International Conference on Energy, Communication, Data Analytics and Soft Computing (ICECDS), Chennai, India, 1–2 August 2017; IEEE: Piscataway, NJ, USA, 2017.
- 88. Nasir, T.; Raza, S.; Abrar, M.; Muqeet, H.A.; Jamil, H.; Qayyum, F.; Cheikhrouhou, O.; Alassery, F.; Hamam, H. Optimal Scheduling of Campus Microgrid Considering the Electric Vehicle Integration in Smart Grid. *Sensors* **2021**, *21*, 7133. [CrossRef] [PubMed]
- 89. Kristiawan, R.B.; Widiastuti, I.; Suharno, S. Technical and economical feasibility analysis of photovoltaic power installation on a university campus in indonesia. *MATEC Web Conf.* **2018**, *197*, 08012. [CrossRef]
- Angelim, J.H.; Affonso, C.M. Energy management on university campus with photovoltaic generation and BESS using simulated annealing. In Proceedings of the 2018 IEEE Texas Power and Energy Conference (TPEC), College Station, TX, USA, 8–9 February 2018; IEEE: Piscataway, NJ, USA, 2018; pp. 1–6.
- Nazir, M.; Patel, T.; Enslin, J.H. Hybrid microgrid controller analysis and design for a campus grid. In Proceedings of the 2019 IEEE 10th International Symposium on Power Electronics for Distributed Generation Systems (PEDG), Xi'an, China, 3–6 June 2019; IEEE: Piscataway, NJ, USA, 2019.

- 92. Savić, N.S.; Katić, V.A.; Katić, N.A.; Dumnić, B.; Milićević, D.; Čorba, Z. Techno-economic and environmental analysis of a microgrid concept in the university campus. In Proceedings of the 2018 International Symposium on Industrial Electronics (INDEL), Banja Luka, Bosnia and Herzegovina, 1–3 November 2019; IEEE: Piscataway, NJ, USA, 2019. [CrossRef]
- 93. Muqeet, H.A.U.; Ahmad, A. Optimal Scheduling for Campus Prosumer Microgrid Considering Price Based Demand Response. IEEE Access 2020, 8, 71378–71394. [CrossRef]
- 94. Liu, Z.; Chen, C.; Yuan, J. Hybrid Energy Scheduling in a Renewable Micro Grid. Appl. Sci. 2015, 5, 516–531. [CrossRef]
- 95. Washom, B.; Dilliot, J.; Weil, D.; Kleissl, J.; Balac, N.; Torre, W.; Richter, C. Ivory Tower of Power: Microgrid Implementation at the University of California, San Diego. *IEEE Power Energy Mag.* **2013**, *11*, 28–32. [CrossRef]
- Bracco, S.; Delfino, F.; Pampararo, F.; Robba, M.; Rossi, M. Planning and management of sustainable microgrids: The test-bed facilities at the University of Genoa. In Proceedings of the 2013 Africon, Pointe-Aux-Piments, Mauritius, 9–12 September 2013; pp. 1–5.
- 97. Setthapun, W.; Srikaew, S.; Rakwichian, J.; Tantranont, N.; Rakwichian, W.; Singh, R. The integration and transition to a DC based community: A case study of the Smart Community in Chiang Mai World Green City. In Proceedings of the 2015 IEEE First International Conference on DC Microgrids (ICDCM), Atlanta, GA, USA, 7–10 June 2015; IEEE: Piscataway, NJ, USA, 2015; pp. 205–209.
- Symposiums, I.M. Hangzhou Dianzi University. 2007. Available online: https://microgrid-symposiums.org/microgridexamples-and-demonstrations/hangzhou-dianzi-university-microgrid/ (accessed on 9 January 2024).
- Mihet-Popa, L.; Groza, V.; Isleifsson, F. Experimental testing for stability analysis of distributed energy resources components with storage devices and loads. In Proceedings of the 2012 IEEE International Instrumentation and Measurement Technology Conference (I2MTC), Graz, Austria, 13–16 May 2012; IEEE: Piscataway, NJ, USA, 2012; pp. 588–593.
- Ogbikaya, S.; Iqbal, M.T. Design and sizing of a microgrid system for a University community in Nigeria. In Proceedings of the 2022 IEEE 12th Annual Computing and Communication Workshop and Conference (CCWC), Las Vegas, NV, USA, 26–29 January 2022; IEEE: Piscataway, NJ, USA, 2022.
- Bin, L.; Shahzad, M.; Javed, H.; Muqeet, H.A.; Akhter, M.N.; Liaqat, R.; Hussain, M.M. Scheduling and Sizing of Campus Microgrid Considering Demand Response and Economic Analysis. *Sensors* 2022, 22, 6150. [CrossRef]
- 102. Khodayar, M.E.; Barati, M.; Shahidehpour, M. Integration of High Reliability Distribution System in Microgrid Operation. *IEEE Trans. Smart Grid* 2012, *3*, 1997–2006. [CrossRef]
- Lofberg, J. YALMIP: A toolbox for modeling and optimization in MATLAB. In Proceedings of the 2004 IEEE international conference on robotics and automation (IEEE Cat. No. 04CH37508), Taipei, Taiwan, 2–4 September 2004; IEEE: Piscataway, NJ, USA, 2004.
- Fu, L.; Meng, K.; Liu, B.; Dong, Z.Y. Mixed-integer second-order cone programming framework for optimal scheduling of microgrids considering power flow constraints. *IET Renew. Power Gener.* 2019, 13, 2673–2683. [CrossRef]
- 105. Esmaeili, Z.; Hosseini, S.H. Optimal Energy Management of Microgrids using Quantum Teaching Learning Based Algorithm. *AUT J. Electr. Eng.* **2023**, *55*, 8.
- 106. Li, Y.; Li, L.; Jiang, Y.; Gan, Y.; Zhang, J.; Yuan, S. A Novel Hybrid Maximum Power Point Tracking Technique for PV System under Complex Partial Shading Conditions in Campus Microgrid. *Appl. Sci.* **2023**, *13*, 4998. [CrossRef]
- Rajagopalan, A.; Nagarajan, K.; Montoya, O.D.; Dhanasekaran, S.; Kareem, I.A.; Perumal, A.S.; Lakshmaiya, N. Paramasivam, Multi-Objective Optimal Scheduling of a Microgrid Using Oppositional Gradient-Based Grey Wolf Optimizer. *Energies* 2022, 15, 9024. [CrossRef]
- 108. Sardou, I.G.; Zare, M.; Azad-Farsani, E. Robust energy management of a microgrid with photovoltaic inverters in VAR compensation mode. *Int. J. Electr. Power Energy Syst.* 2018, *98*, 118–132. [CrossRef]
- 109. Jaramillo, L.B.; Weidlich, A. Optimal microgrid scheduling with peak load reduction involving an electrolyzer and flexible loads. *Appl. Energy* **2016**, *169*, 857–865. [CrossRef]
- 110. Guo, Y.; Sheng, S.; Anglani, N.; Lehman, B. Economically Optimal Power Flow Management of Grid-Connected Photovoltaic Microgrid Based on Dynamic Programming Algorithm and Grid I/O Strategy for Different Weather Scenarios. In Proceedings of the 2019 IEEE Applied Power Electronics Conference and Exposition (APEC), Anaheim, CA, USA, 17–21 March 2019; IEEE: Piscataway, NJ, USA, 2019; pp. 174–181.
- 111. Rullo, P.; Braccia, L.; Luppi, P.; Zumoffen, D.; Feroldi, D. Integration of sizing and energy management based on economic predictive control for standalone hybrid renewable energy systems. *Renew. Energy* **2019**, *140*, 436–451. [CrossRef]
- 112. Indragandhi, V.; Logesh, R.; Subramaniyaswamy, V.; Vijayakumar, V.; Siarry, P.; Uden, L. Multi-objective optimization and energy management in renewable based AC/DC microgrid. *Comput. Electr. Eng.* **2018**, *70*, 179–198. [CrossRef]
- Mellouk, L.; Ghazi, M.; Aaroud, A.; Boulmalf, M.; Benhaddou, D.; Zine-Dine, K. Design and energy management optimization for hybrid renewable energy system- case study: Laayoune region. *Renew. Energy* 2019, 139, 621–634. [CrossRef]
- 114. Ndwali, K.; Njiri, J.G.; Wanjiru, E.M. Multi-objective optimal sizing of grid connected photovoltaic batteryless system minimizing the total life cycle cost and the grid energy. *Renew. Energy* **2020**, *148*, 1256–1265. [CrossRef]
- 115. Niyomubyeyi, O.; Sicuaio, T.E.; González, J.I.D.; Pilesjö, P.; Mansourian, A. A Comparative Study of Four Metaheuristic Algorithms, AMOSA, MOABC, MSPSO, and NSGA-II for Evacuation Planning. *Algorithms* **2020**, *13*, 16. [CrossRef]
- 116. Helmi, A.M.; Carli, R.; Dotoli, M.; Ramadan, H.S. Efficient and Sustainable Reconfiguration of Distribution Networks via Metaheuristic Optimization. *IEEE Trans. Autom. Sci. Eng.* **2021**, *19*, 82–98. [CrossRef]

- 117. Güven, A.F.; Yörükeren, N.; Tag-Eldin, E.; Samy, M.M. Multi-Objective Optimization of an Islanded Green Energy System Utilizing Sophisticated Hybrid Metaheuristic Approach. *IEEE Access* **2023**, *11*, 103044–103068. [CrossRef]
- 118. Hertzog, P.E.; Swart, A.J. A Case Study: Validating a LabVIEW simulation models presentation of fundamental principles in emerging renewable energy systems. In Proceedings of the 2016 IEEE International Conference on Emerging Technologies and Innovative Business Practices for the Transformation of Societies (EmergiTech), Balaclava, Mauritius, 3–6 August 2016; IEEE: Piscataway, NJ, USA, 2016.
- 119. Obeng, M.; Gyamfi, S.; Derkyi, N.S.; Kabo-Bah, A.T.; Peprah, F. Technical and economic feasibility of a 50 MW grid-connected solar PV at UENR Nsoatre Campus. *J. Clean. Prod.* 2020, 247, 119159. [CrossRef]
- Thotakura, S.; Kondamudi, C.; Xavier, J.F.; Quanjin, M.M.; Reddy, G.R.; Gangwar, P.; Davuluri, L. Operational performance of megawatt-scale grid integrated rooftop solar PV system in tropical wet and dry climates of India. *Case Stud. Therm. Eng.* 2020, 18, 100602. [CrossRef]
- 121. Kumar, N.M.; Sudhakar, K.; Samykano, M. Techno-economic analysis of 1 MWp grid connected solar PV plant in Malaysia. *Int. J. Ambient. Energy* **2019**, *40*, 434–443. [CrossRef]
- 122. Alwazani, H.; Bahanshal, S.; Kumar, C.R.; Majid, M.; Brahma, J.; Kanneganti, V.; Wadgaonkar, A.; Kurian, M.; Rehman, A.; Bugshan, H. Economic and Technical Feasibility of Solar System at Effat University. In Proceedings of the 2019 IEEE 10th GCC Conference & Exhibition (GCC), Kuwait City, Kuwait, 19–23 April 2019; IEEE: Piscataway, NJ, USA, 2019; pp. 1–5.
- O'Donovan and Mohamed Al-Musleh. Heriot-Watt University Launches Dubai Solar Test Site for Companies in UK and Beyond. 2023. Available online: https://www.hw.ac.uk/news/articles/2023/heriot-watt-university-launches-dubai-solar.htm (accessed on 7 January 2023).
- 124. Solar Power Generation Facilities on Campus. Available online: https://www.cuc.ac.jp/eng/energy/solar/index.html (accessed on 22 November 2023).
- 125. Al-Abed, M. Renewable Energy Initiatives at The Hashemite University. 2023. Available online: https://www.ecomena.org/ renewable-energy-hashemite-university/ (accessed on 29 January 2024).
- Hartley, M. Rooftop Tour Shows Off the University's New Solar Array. 2022. Available online: https://www.albany.edu/news-center/news/2022-rooftop-tour-shows-universitys-new-solar-array (accessed on 29 January 2024).
- 127. AKU Initiates First Solar Power Project on the Campus. 2022. Available online: https://www.aku.edu/news/Pages/News\_ Details.aspx?nid=NEWS-002685 (accessed on 3 January 2024).
- 128. Sylte, A. Photos: CSU Celebrates Nearly Doubling Number of Solar Installations on Campus. 2022. Available online: https://source.colostate.edu/csu-new-solar-panels/ (accessed on 13 January 2024).
- 129. UNC and McKinstry Installing Solar Array on Campus. 2019. Available online: https://www.unco.edu/news/newsroom/ releases/unc-mckinstry-solar-array-installation.aspx (accessed on 30 January 2024).
- 130. Goers, A. UW-Stout's Largest Solar Panel Array to Date Installed, Boosting Campus Sustainability. 2023. Available online: https://www.wisconsin.edu/all-in-wisconsin/story/uw-stouts-largest-solar-panel-array-to-date-installed-boosting-campussustainability/ (accessed on 31 January 2024).
- Strawn, J. IE Alums Add Solar Power to Campus. 2015. Available online: https://news.engineering.iastate.edu/2015/09/28/iealums-add-solar-power-to-campus/ (accessed on 31 January 2024).
- 132. Solar Panels for Universities. Available online: https://www.solarsense-uk.com/filter\_sector/universities/ (accessed on 31 January 2024).
- 133. Nair, L. W&L to Match 100% of Electricity Use with Solar Energy. 2022. Available online: https://columns.wlu.edu/wl-to-match-100-of-electricity-use-with-solar-energy/ (accessed on 1 February 2024).
- 134. FREDERIC BROWN. University of Texas Develops New Solar System Integrating Energy Generation and Storage. 2018. Available online: https://www.pv-magazine.com/2018/05/24/university-of-texas-develops-new-solar-system-integrating-energy-generation-and-storage/ (accessed on 30 January 2024).
- 135. Khan, M.R.B.; Pasupuleti, J.; Al-Fattah, J.; Tahmasebi, M. Optimal Grid-Connected PV System for a Campus Microgrid. *Indones. J. Electr. Eng. Comput. Sci.* 2018, 12, 899–906. [CrossRef]
- Haffaf, A.; Lakdja, F.; Abdeslam, D.O. Experimental performance analysis of an installed microgrid-based PV/battery/EV grid-connected system. *Clean Energy* 2022, 6, 599–618. [CrossRef]
- 137. Solar Panels and Energy Storage System Make Debut on Largest Building on Campus. 2016. Available online: https://www.northcentralcollege.edu/news/2016/12/09/solar-panels-and-energy-storage-system-make-debut-largest-building-campus (accessed on 30 January 2024).
- VALLEY, S. Gavilan Develops Solar and Storage Project. 2017. Available online: https://sanbenito.com/gavilan-develops-solarstorage-project/ (accessed on 30 January 2024).
- Desk, E. EDF Renewables North America to Provide Battery Energy Storage, Onsite Solar and EV Charging Stations for Sanford Burnham Prebys Campus. 2020. Available online: https://solarquarter.com/2020/07/23/edf-renewables-north-america-toprovide-battery-energy-storage-onsite-solar-and-ev-charging-stations-for-sanford-burnham-prebys-campus/ (accessed on 30 January 2024).
- 140. Bachman, J. NMSU, El Paso Electric Solar Installation Begins Generating Power. 2022. Available online: https://www. lascrucesbulletin.com/stories/nmsu-el-paso-electric-solar-installation-begins-generating-power,11992 (accessed on 30 January 2024).

- 141. Hall, H. Yotta to Install SolarLEAF at California University. 2020. Available online: https://www.rdworldonline.com/yotta-to-install-solarleaf-at-california-university/ (accessed on 30 January 2024).
- 142. National Renewable Energy Laboratory. Available online: https://www.nrel.gov/docs/fy17osti/70037.pdf (accessed on 2 February 2024).
- 143. National Renewable Energy Laboratory. How To Afford Investment in Clean Energy. Available online: https://www.nrel.gov/ wind/ (accessed on 15 February 2024).
- 144. Green Initiatives—Wind Garden & Photovoltaic Solar Panels Quinnipiac University (York Hill Campus), Hamden, Connecticut. Available online: https://www.pegasusgrp.net/green-initiatives--wind-garden--photovoltaic-solar-panels---quinnipiac-u.html (accessed on 31 January 2024).
- 145. Low Wind Speed Case Study University of Delaware Wind Turbine. 2014. Available online: https://cleanenergy.org/wp-content/uploads/UD-Elevated-Opportunities-Wind-Technology-for-the-South.pdf (accessed on 22 January 2024).
- 146. Marquart, C. Wind Over Morris. 2021. Available online: https://wcroc.cfans.umn.edu/ (accessed on 31 January 2024).
- 147. College, S.O.; Wind at St. Olaf. Available online: https://wp.stolaf.edu/facilities/wind-at-st-olaf/ (accessed on 1 February 2024).
- Macalester College, St. Paul, MN: Community Wind Project. 2003. Available online: https://www.windustry.org/resources/ macalester-college-st-paul-mn-community-wind-project (accessed on 1 February 2024).
- 149. Hybrid Energy, PV & Storage Systems for Integrated Power Plants. Available online: https://eps.fiu.edu/hybrid-energy-storagesystems-for-distributed-energy-resources/ (accessed on 1 February 2024).
- 150. Maritz, J. Optimized Energy Management Strategies for Campus Hybrid PV–Diesel Systems during Utility Load Shedding Events. *Processes* 2019, 7, 430. [CrossRef]
- 151. Ndwali, P.K.; Njiri, J.G.; Wanjiru, E.M. Optimal Operation Control of Microgrid Connected Photovoltaic-Diesel Generator Backup System Under Time of Use Tariff. *J. Control Autom. Electr. Syst.* **2020**, *31*, 1001–1014. [CrossRef]
- Combined Heat and Power. Available online: https://www.energy.ca.gov/data-reports/california-power-generation-and-power-sources/combined-heat-and-power (accessed on 15 February 2024).
- 153. Zhang, G.; Cao, Y.; Cao, Y.; Li, D.; Wang, L. Optimal Energy Management for Microgrids with Combined Heat and Power (CHP) Generation, Energy Storages, and Renewable Energy Sources. *Energies* **2017**, *10*, 1288. [CrossRef]
- 154. Testi, D.; Conti, P.; Schito, E.; Urbanucci, L.; D'ettorre, F. Synthesis and Optimal Operation of Smart Microgrids Serving a Cluster of Buildings on a Campus with Centralized and Distributed Hybrid Renewable Energy Units. *Energies* 2019, 12, 745. [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.